

Working Paper Series

Bank Privatization and Productivity: Evidence for Brazil Márcio I. Nakane and Daniela B. Weintraub December, 2004

CGC 00.030.100/0001-0					
Working Paper Series	Brasília	n. 90	Dec	2004	P. 1-54

ISSN 1518-3548 CGC 00.038.166/0001-05

Working Paper Series

Edited by Research Department (Depep) - E-mail: workingpaper@bcb.gov.br

Editor: Benjamin Miranda Tabak – E-mail: benjamin.tabak@bcb.gov.br Editorial Assistent: Jane Sofia Moita – E-mail: jane.sofia@bcb.gov.br Head of Research Department: Marcelo Kfoury Muinhos – E-mail: marcelo.kfoury@bcb.gov.br

The Banco Central do Brasil Working Papers are all evaluated in double blind referee process.

Reproduction is permitted only if source is stated as follows: Working Paper n. 90.

Authorized by Afonso Sant'Anna Bevilaqua, Deputy Governor of Economic Policy.

General Control of Publications

Banco Central do Brasil Secre/Surel/Dimep SBS – Quadra 3 – Bloco B – Edifício-Sede – M1 Caixa Postal 8.670 70074-900 Brasília – DF – Brazil Phones: (5561) 414-3710 and 414-3567 Fax: (5561) 414-3626 E-mail: editor@bcb.gov.br

The views expressed in this work are those of the authors and do not necessarily reflect those of the Banco Central or its members.

Although these Working Papers often represent preliminary work, citation of source is required when used or reproduced.

As opiniões expressas neste trabalho são exclusivamente do(s) autor(es) e não refletem, necessariamente, a visão do Banco Central do Brasil.

Ainda que este artigo represente trabalho preliminar, citação da fonte é requerida mesmo quando reproduzido parcialmente.

Banco Central do Brasil Information Bureau

Address:	Secre/Surel/Diate
	SBS – Quadra 3 – Bloco B
	Edifício-Sede – 2º subsolo
	70074-900 Brasília – DF – Brazil
Phones:	(5561) 414 () 2401, 2402, 2403, 2404, 2405, 2406
DDG:	0800 992345
Fax:	(5561) 321-9453
Internet:	http://www.bcb.gov.br
E-mails:	cap.secre@bcb.gov.br
	dinfo.secre@bcb.gov.br

Bank Privatization and Productivity: Evidence for Brazil^{*}

Márcio I. Nakane**

Daniela B. Weintraub***

Abstract

Over the last decade, the Brazilian banking industry has undergone major and deep transformations with several privatizations of state-owned banks, mergers and acquisitions, closing down of troubled banks, entry by foreign banks, etc. The purpose of this paper is to evaluate the impacts of these changes in banking total factor productivity. We first obtain measures of bank level productivity by employing the techniques due to Levinsohn and Petrin (2003). We then relate such measures to a set of bank characteristics. Our main results indicate that state-owned banks are less productive than their private peers, and that privatization has increased productivity.

Keywords: bank privatization, total factor productivity, Brazilian banking JEL Classification: G21, L33, O54

^{*} Comments from Leonardo S. Alencar, Renato Baumann, Simão Silber, José Carlos Souza Santos, Naércio Menezes, Eduardo L. Lundberg, Victorio Chu, Francisco Pérez-González, and seminar participants at the Central Bank of Brazil, University of Wales at Bangor, Federal University of Ceará, University of São Paulo, Ibmec, the IX CEMLA Meeting of the Network of Central Bank Researchers, and the 9th Annual LACEA Meeting are acknowledged. Adriana Schor helped with the computer routines. Thorsten Beck kindly sent us some bank information. Remaining errors are our own responsibility. The views expressed here are solely the responsibility of the authors and do not reflect those of the Central Bank of Brazil or its members.

^{**} Central Bank of Brazil, Research Department and University of São Paulo. E-mail: marcio.nakane@bcb.gov.br.

^{****} University of São Paulo. E-mail: dbaumohl@uol.com.br.

1. Introduction

The banking system in Brazil is the largest and the most complex one in Latin America. Like in many parts of the world, the banking industry in Brazil is undergoing a process of rapid and radical transformations. The common features of this process, in Brazil and elsewhere, include: an increase in competition from within the industry as well as from the outside; a wave of merger and acquisition (M&A) activities, including several cross-border deals; more globalized capital markets with highly volatile capital flows, which are capable of causing havoc in some national financial sectors; new financial products, with increasing reliance on off-balance sheet activities; new banking practices brought out by the information technology revolution.

The banking sector in Brazil has been strongly influenced by the changing domestic macroeconomic scene of the recent period, especially by the transition from a high to a low inflation environment. After many years of making a living out of inflationary rents, this transition was far from smooth for many banks.

Amongst those most affected by the many changes in the industry were the stateowned banks. Due to their poor performance, many of the state-owned banks in Brazil were either closed down or privatized. Less than half (14) of the 32 state-owned banks operating in the country by 1994 were still active by 2002.

The Brazilian experience represents an interesting case study on bank privatization not only because of its quantitative relevance but also due to the varied options given to the state-owned banks following their restructuring. Thus, some stateowned banks were straight privatized by their controllers (namely, the Brazilian states) whereas some others had their control first transferred from the states to the federal government and then privatized. Some other states also kept the control of their banks after restructuring. There are also some other state-owned banks that were just liquidated.

The purpose of this paper is to study the impact of the privatization of stateowned banks on productivity. Measures of bank-level total factor productivity are first obtained as the residuals from a production function estimate. The production function is estimated following the methodology suggested by Levinsohn and Petrin (2003) to try to control for endogeneity problems arising from the simultaneous choice of inputs and productivity by the bank firm.

In a second stage, bank total factor productivity is related to a set of control variables. In an environment where many different types of corporate control changes are occurring simultaneously, it is important to try to control for as many of them as possible even though the primary interest of the paper lies on the effects of bank privatization. This is certainly the case for Brazil whereby privatization of state-owned banks were taking place alongside other corporate changes in the industry like domestic mergers and acquisitions, foreign acquisition of domestic banks, liquidation of banks, and pure exit from the market. We therefore follow the methodology proposed by Berger et al. (2003) and include variables controlling for static, selection, and dynamic effects. Static variables are dummies for groups of banks that have not had any corporate change over the sample period. Selection variables are dummies for groups of banks that have had some corporate change over the sample period. Such dummy variables are equal to one over the *whole* sample period for the corresponding banks. Dynamic variables are of two forms. A first set of dynamic variables are dummies for those banks that have had some corporate change over the sample period taking the value one only for the time periods following the change. A second set of dynamic variables track the number of time periods following the change. In addition to the static, selection, and dynamic variables we also included a set of dummy variables for those banks that have exited the market. Exit can occur either because the bank has been liquidated or because the bank has changed the nature of its activities.

Our main results show that state-owned banks are less productive than privateowned ones. Another main result of the paper is that privatization has had a positive impact on productivity. Moreover, the positive effects of privatization seem to take some time to materialize. Privatization proved also to be a superior strategy than restructuring and keeping the bank under state control. On the other hand, we could not find any strong performance differences related to the way a state-owned bank was privatized (i.e., straight privatization or federalization followed by privatization). We find no strong differences in the performances of the state-owned banks sold to foreign ones *vis a vis* state-owned banks sold to domestic ones either. This paper contributes to the literature on bank productivity. The study of bank productivity is relevant because productivity is a summary performance measure. Thus, productivity analysis may be relevant to those involved in bank M&A issues, like bank practitioners or bank competition authorities. Also, to the extent that low productivity can work as an early warning, bank supervision authorities may use productivity measures as an additional monitoring instrument. Bank productivity studies are also useful due to the well-documented evidence that a bank system that efficiently channels available resources to productive uses is a powerful mechanism for economic growth [Levine (1997)].

This paper is also related to the literature on bank privatization. The empirical literature in this area takes the form of either cross-country studies or analyses of individual countries.¹ This literature provides broad support to the conclusions reached in this paper on the poorer performance of state-owned banks and on the beneficial impacts of bank privatization.

This paper is structured as follows. Section 2 gives a brief overview of the banking industry in Brazil, with a special emphasis on the state-owned sector. Section 3 describes the methodology to be applied in the empirical sections. Section 4 discusses data-related issues. Section 5 estimates the coefficients of a production function, from which the bank-level productivity measures are calculated. Section 6 studies the determinants of bank productivity, highlighting the role played by bank privatization. Section 7 concludes the paper.

2. Overview of the banking sector in Brazil

The launching of the stabilization plan, called the *Real Plan*, in July 1994 with the subsequent transition to an environment of more stable prices proved to be very costly to the Brazilian banks. During the high-inflation period, banks could profit from inflation transfers. Inflation imposes a tax over the holders of money and non-interest

¹ Boehmer *et al.* (2003), Bonin *et al.* (2003), Otchere (2003), and Nguyen and Williams (2003) are examples of cross-country studies. Some studies focusing on individual countries are, among others, Berger *et al.* (2003) for Argentina, Beck, Crivelli, and Summerhill (2003) for Brazil, Beck, Cull, and Jerome (2003) for Nigeria, Haber and Kantor (2003) for Mexico, Omran (2003) for Egypt, and

bearing deposits. As issuers of demand deposits, commercial banks receive part of the inflation tax. According to ANDIMA-IBGE (1997), the inflationary transfers to the banking system fell from an average of 3.4% of GDP in the 1990-93 period to 1.8% in 1994, and to 0.03% in 1995.

In the immediate aftermath of the stabilization plan, Brazilian banks tried to make up for the inflationary losses by increasing credit. Total loans of the financial system went up 43.7% after the first eight months of the stabilization plan. The rapid increase in the concession of loans was not followed by a careful consideration of the risk characteristics of those seeking credit. When the Central Bank dramatically increased the reserve requirements on deposits in the second half of 1994, coupled with the continuation of a policy of high interest rates, a credit retrenchment followed. Non-performing loans started to accumulate fast.

With the imminent insolvency of some big private banks² a bailout mechanism was put in place in November 1995³. Under this program, the Central Bank was given the mandate to compel a fragile bank to: a) increase its capital, or b) to transfer its shareholder control, or c) to be merged or acquired by another bank. PROER made easier for stronger financial institutions to acquire weaker ones by allowing the acquiring financial institutions to record as a premium the difference between the acquisition value and the market value of the acquired institution. Non-performing loans were recognized as losses and, under certain conditions, the premium could be used as a tax credit. It also allowed forbearance in the form of a temporary waive of the Basle minimum capital requirement for the ailing participants. In order to reduce the moral hazard problems associated to bailout schemes, PROER set out that banks could only qualify for official help when the ownership control was agreed to be transferred to some other institution. Seven banking institutions were restructured under the PROER resources.

Bonaccorsi di Patti and Hardy (2003) for Pakistan. Megginson (2003) and Clarke *et al.* (2003) provide comprehensive surveys about bank privatization studies.

² In August 1995, Banco Economico, the eight largest in the country by net worth, fell under Central Bank intervention. In November 1995, the same fate hit Banco Nacional, the sixth largest in the country by net worth.

³ Program of Incentives to the Restructuring and Strengthening of the National Financial System, PROER.

The PROER program only reached private banks. A similar program aiming at the state-owned banks was launched in August 1996, the PROES⁴. The aim of this program was not only to reduce the participation of the Brazilian states (provinces) in the banking activity but also to address their chronic public debt problems⁵. Debt restructuring packages were offered for those states who agreed to give one of the following destinations to their banks: a) to liquidate it; b) to privatize it; c) to transfer its control to the federal government for future privatization, or, d) to transform it in a development agency. Less favorable financial packages were also offered to those states that still wanted to keep their banks after bailing them out.

When PROES was launched, there were 35 financial institutions under the control of the Brazilian states, including 23 commercial banks. With the exception of the states of Mato Grosso do Sul, Tocantins (which had no financial institutions under their control), Paraíba, and Distrito Federal, all the other state units joined PROES. Under PROES, ten financial institutions were/are being liquidated, six banks were privatized by the states, six banks were/are being privatized by the federal government, sixteen financial institutions were transformed into development agencies, and five banks were restructured and kept under the state control⁶.

PROES only reached the banks owned by the states. For the banks owned by the federal government, an official restructuring program was launched in June 2001, the PROEF⁷. Under this program, many troubled assets were transferred to a newly created institution under the Finance Ministry control. Three federal government-owned banks (CEF, BNB, and Basa) also received capital injection⁸.

The whole set of measures put in place drastically changed the ownership composition of the banking sector in the country. Tables 1 to 5 document such changes along several dimensions.

⁴ Program of Incentives to the Reduction of the State-Level Public Sector in the Bank Activity. See also Baer and Nazmi (2000), and Ness Jr. (2000) for more details.

⁵ The two problems were not unrelated: state-level banks were the main purchasers of the public bonds issued by their main shareholders, the states themselves. Werlang and Fraga Neto (1995) study the role of state-owned banks in the creation of public debt. Bevilaqua (2000) describes state debt developments from the mid-1980s onwards, with special emphasis on the 1997 state debt bailout.

⁶ See Appendix 1 for a list of the liquidated and privatized banks.

⁷ Program for the Strengthening of the Federal Financial Institutions.

⁸ Banco do Brasil (BB), another federal government-owned bank, was capitalized by the Treasury back in 1995.

	Private	Foreign	Foreign	State	
Year	Domestic	Minority	Control	Owned	TOTAL
1994	146	31	37	32	246
1995	142	32	36	32	242
1996	130	29	40	32	231
1997	119	26	45	27	217
1998	105	17	58	23	203
1999	96	12	67	19	194
2000	93	13	69	17	192
2001	82	14	70	16	182
2002	75	10	56	14	155

TABLE 1 Number of Commercial Banks by Ownership

The number of commercial banks operating in Brazil has been reducing since 1994. There were 91 fewer banks working in the country in 2002 than in 1994. Apart from foreign controlled banks, all the other bank segments showed considerable reductions in their numbers. Even for foreign controlled banks, the year 2002 represented a reversal of the trend when, after many years of continuing expansion, their number showed a reduction.

	Private	Control	State
Year	Domestic	Foreign	Owned
1993	48.47	7.32	44.21
1994	56.03	9.64	34.33
1995	49.93	13.27	36.80
1996	56.03	10.42	33.55
1997	52.64	14.51	32.85
1998	50.55	22.21	27.23
1999	47.55	25.93	26.53
2000	51.35	28.88	19.76
2001	52.19	31.35	16.45
2002	49.74	33.62	16.64

TABLE 2: Share of bank system net worth by ownership

TABLE 3: Share of bank system assets by ownership

	Private	Foreign	State
Year	Domestic	Control	Owned
1993	40.72	8.36	50.92
1994	41.29	7.17	51.53
1995	39.25	8.41	52.34
1996	39.12	9.82	51.06
1997	36.90	12.87	50.24
1998	35.47	18.47	46.06
1999	33.33	23.34	43.33
2000	35.50	27.62	36.88
2001	37.55	30.13	32.32
2002	37.32	27.67	35.01

	Private	Foreign	State
Year	Domestic	Control	Owned
1993	38.85	4.84	56.32
1994	39.41	4.59	56.00
1995	36.48	5.41	58.11
1996	34.18	4.37	61.45
1997	33.01	7.58	59.42
1998	33.28	15.23	51.49
1999	32.07	16.93	50.99
2000	34.28	21.36	44.36
2001	35.80	20.41	43.79
2002	37.16	20.13	42.71

 TABLE 4: Share of bank system deposits by ownership

TABLE 5: Share of bank system loans by ownership

	Private	Foreign	State
Year	Domestic	Control	Owned
1993	31.55	6.57	61.88
1994	35.47	5.20	59.34
1995	31.93	5.75	62.32
1996	32.91	8.70	58.39
1997	35.61	11.80	52.59
1998	31.26	15.02	53.72
1999	32.03	19.98	48.00
2000	34.96	25.48	39.56
2001	42.82	32.03	25.15
2002	40.45	30.48	29.07

Tables 2 to 5^9 show that private domestic banks managed to keep their share of the bank system net worth, assets, and deposits. This group of banks even increased their share of the bank system loans.

Foreign controlled banks increased significantly their market penetration in the country. In 2002, they accounted for 33.6% of the bank system net worth, 27.7% of the bank system assets, 20.1% of the bank system deposits, and 30.5% of the bank system loans.

Despite the great reduction in the importance of the state-owned banks, they still account for significant shares of the bank system net worth (16.6%), assets (35%), deposits (42.7%), and loans (29.1%).

3. Methodology

Productivity is defined as any variation in output that cannot be explained by variations in inputs. On this account, productivity changes can be due either to variations in efficiency or to changes in technology.

The methodology to be applied in this paper follows the approach developed by Olley and Pakes (1996). A panel data for Brazilian banks will be used with the aim of estimating the parameters of a production function. In usual fashion, productivity is measured as the residual from this relation.

Olley and Pakes' approach allows one to consistently estimate the production function coefficients taking into consideration two possible sources of bias, namely a sample selection and a simultaneity bias.

The sample selection bias refers to the fact that many banks left the market during the sample period. It is reasonable to imagine that the unobservable productivity variable and the decision to leave the market are correlated, causing a potential sample selection problem. The simultaneity problem is related to the correlation between the unobservable productivity variable and the amount of inputs chosen by the bank.

⁹ In tables 2 to 5, the foreign minority group is incorporated into the private domestic group.

The authors deal with both problems by first modeling the optimal firm decision regarding both the use of inputs as well as the market exit/no exit. The solution to the firm control problem takes the form of an exit rule and an investment demand function, respectively given by:

$$\chi_{t} = \begin{cases} 1 & \text{if } \omega_{t} \ge \underline{\omega}_{t}(k_{t}) \\ 0 & \text{otherwise} \end{cases}$$
(1)

$$i_t = i_t(\omega_t, k_t) \tag{2}$$

where χ_t is an indicator function equal to zero when the firm exits the market, ω_t is the firm productivity index, k_t is the firm capital stock, i_t is the firm investment level, and $\underline{\omega}_t(.)$ and $i_t(.)$ are functions that are determined as part of the Nash Markov perfect equilibrium of the firm optimization problem.

The procedure to be adopted in the estimation can be illustrated taking the example of a Cobb-Douglas production function:

$$y_{it} = \beta_0 + \beta_k k_{it} + \beta_l l_{it} + \omega_{it} + \eta_i$$
(3)

where y_{it} is the log of the output of firm *i* in period *t*, k_{it} is the log of its capital stock, l_{it} is the log of its labor input, ω_{it} is its productivity, and η_{it} is an error stochastic term. Notice that both ω and η are not observed by the econometrician. The difference between them is that ω is a state variable in the firm decision problem and therefore it affects both the exit decision as well as the demand for inputs, while η has no such implications.

Olley and Pakes propose a three-step procedure to estimate the coefficients of (3) taking into consideration the sample selection and the simultaneity problems. In order to implement the first step, the inversion of (2), which is strictly increasing in ω , allows one to write:

$$\boldsymbol{\omega}_{t} = \boldsymbol{h}_{t}(\boldsymbol{i}_{t},\boldsymbol{k}_{t}) \tag{4}$$

Equation (4) expresses the unobserved productivity variable as a function of observable variables. By replacing (4) in (3) it is possible therefore to control for ω in the estimation:

$$y_{it} = \beta_l l_{it} + \phi_t (i_{it}, k_{it}) + \eta_{it}$$
(5)

where:

$$\phi_t(i_{it}, k_{it}) \equiv \beta_0 + \beta_k k_{it} + h_t(i_{it}, k_{it})$$
(6)

The "partially linear" model (5)-(6) is a semiparametric regression model. The first step in the estimation allows the identification of the variable input coefficient β_l , but it does not allow the identification of the fixed input coefficient β_k . That is, the estimation of (5)-(6) does not allow one to separate the effect of capital on output from the effect on the investment decision.

The second step involves the determination of the survival probability. After replacing (4) in (1), it is possible to express the survival probability as a function of observable variables only:

$$\operatorname{Prob}\left[\chi_{t+1} = 1 \middle| \underline{\omega}_{t+1}(k_{t+1}), J_t\right] = P_t(i_t, k_t) \equiv P_t$$
(7)

where J_t is the information set available at *t*. Equation (7) can be alternatively estimated by probit/logit, where polynomial terms of i_t and k_t are used as regressors, or by non parametric methods.

The third step in the estimation begins with the assumption that productivity follows a first-order Markov process:

$$\omega_{t+1} = E[\omega_{t+1} | \omega_t, \chi_{t+1} = 1] + \xi_{t+1} = g(\underline{\omega}_{t+1}, \omega_t) + \xi_{t+1} = g(P_t, \phi_t - \beta_k k_t) + \xi_{t+1}$$
(8)

where ξ_{t+1} is the innovation in ω_{t+1} .

By making use of the previous results, it is possible to show that $y_{t+1} - \beta_l l_{t+1}$ can be written in regression form as:

$$y_{t+1} - \beta_l l_{t+1} = \beta_k k_{t+1} + g(P_t, \phi_t - \beta_k k_t) + \xi_{t+1} + \eta_{t+1}$$
(9)

Since it is assumed that the capital stock in a given period is known at the beginning of the period, it then follows that ξ_{t+1} and k_{t+1} are independent variables, which makes possible the consistent estimation of β_k in (9). However, since ξ_{t+1} and l_{t+1} are not independent variables, one can see the need for the first step estimation.

Levinsohn and Petrin (2003) introduced an important improvement in the Olley and Pakes' methodology by making use of an intermediate input instead of investment as a proxy variable for the productivity. Investment can only work as a valid proxy if it does not take zero values, which can be a very restrictive condition for the data sets typically found in developing countries.

Levinsohn and Petrin also argue that the monotonicity condition required for the inversion of (2) may not be valid due to capital adjustment costs. The monotonicity condition for investment is then replaced by an equivalent requirement for the intermediate input function $\tau_t = \tau_t(\omega_t, k_t)$. Another requirement for the use of the intermediate input as a proxy for productivity is an assumption of competitive market. The indexation of the $\tau(.)$ function by *t* allows for temporal changes in prices although variation across firms is not allowed.

The methodology advanced by Levinsohn and Petrin does not require separability of all intermediate inputs in the production function. It is required that only one intermediate input be separable such that its isolated contribution to output can be computed. The authors choose electricity as the productivity proxy on the account that all firms need such input. Moreover, in their sample, there was no firm producing or selling electricity, which can be interpreted as an impossibility for storing such input, making it highly correlated with contemporaneous productivity levels.

The previous equations can then be adapted to the Levinsohn and Petrin extension:

$$\omega_t = h_t(e_t, k_t) \tag{10}$$

$$y_{it} = \beta_l l_{it} + \varphi_t (e_{it}, k_{it}) + \eta_{it}$$
(11)

where e_{it} is electricity, and

$$\varphi_t(e_{it}, k_{it}) \equiv \beta_0 + \beta_e e_{it} + \beta_k k_{it} + h_t(e_{it}, k_{it})$$
(12)

Under the assumption that productivity follows a first-order Markov process, one obtains:

$$\omega_{i_t} = E[\omega_{i_t} | \omega_{i_{t-1}}] = g(\omega_{i_{t-1}}) = g[h_{t-1}(e_{i_{t-1}}, k_{i_{t-1}})]$$
(13)

Lagging (12) one period, replacing in (13), and replacing the result in (11), one obtains:

$$y_{it} - \beta_l l_{it} = \beta_0 + \beta_e e_{it} + \beta_k k_{it} + h_t (e_{it}, k_{it}) + \eta_{it} = \beta_0 + \beta_e e_{it} + \beta_k k_{it} + g [\varphi_{t-1}(e_{it-1}, k_{it-1}) - \beta_0 - \beta_e e_{it-1} - \beta_k k_{it-1}] + \xi_{it} + \eta_{it}$$
(14)

Expression (14) requires the knowledge of φ_{t-1} , which can be estimated in the first-step. Equation (14) is then used to estimate β_e and β_k by non-linear least squares.

In addition to the replacement of investment by an intermediate input as a proxy for productivity, Levinsohn and Petrin do not consider the estimation of the second step in the Olley and Pakes algorithm. In other terms, they do not account for the sample selection bias. Levinsohn and Petrin argue that the use of an unbalanced panel controls, to some extent, for such bias. Moreover, both Olley and Pakes (1996) and Levinsohn and Petrin (1999) found that controlling for selection has little effect on the final parameter estimates.

4. Data and sample

The empirical section of the paper aims at estimating a Cobb-Douglas production function having a measure of bank output as the dependent variable and three productive inputs as explanatory variables. Appendix 2 gives more detailed information on data sources and variable definitions.

The inputs are labor, capital, and intermediate inputs. Following Levinsohn and Petrin (2003), a separate intermediate input is used as a proxy variable for productivity. Unlike the manufacturing sector, however, the use of electricity as a proxy variable in the banking industry does not seem to be warranted. We therefore take communications as our proxy for the unobserved productivity. Capital stock is treated as a fixed input since adjustment costs may prevent instantaneous reallocations of such input. As for labor, we treat it alternatively as a variable and as a fixed factor.¹⁰

Measurement of bank output is more controversial, with many approaches being proposed in the literature¹¹. Here, output is measured as the value of total bank working assets, making our model consistent with the intermediation approach. Haynes and Thompson (1999) use a similar procedure. Some empirical studies use bank deposits either as output or as input of the bank activity. However, because of the relevance of bank deposits in the liability side of a bank balance sheet, the inclusion of them would cause a serious problem in the estimation of the production function, due to the accounting identity equating total assets and total liabilities. In the present study therefore bank deposits are not included in the bank production process.

The source of the accounting data is COSIF (Accounting Plan of the National Financial System Institutions), elaborated by the Brazilian Central Bank, and by which all the financial institutions operating in the country have to report balance sheet and income statements on a monthly basis. The accounts for (end of) June and (end of) December of each year during the period from December 1990 to December 2002 were used.

The sample is unbalanced with 242 commercial banks. All the observations with zero values for the output or for one of the inputs were excluded from the analysis. In addition, banks with less than three observations, and outliers were also excluded. The final sample contains 4,444 observations¹².

¹⁰ Labor economists in Brazil argue that, due to rigid labor market legislation, dismissal of labor force is very costly. Such friction can prevent instantaneous reallocations of the labor input making it behave as a fixed factor.

¹¹ See, among others, Berger and Humphrey (1992), and Fixler and Zieschang (1992).

 $^{^{12}}$ The criterion employed to eliminate the outliers was the following: initially, the ratios of output to labor, of output to capital, and of output to intermediate inputs were computed for all the valid observations. The observations in the lower 0.5% and in the upper 0.5% for each of the three ratios were excluded from the sample.

Table 6 shows some descriptive statistics for the sample variables for both the whole sample period, 1990-2002, and also for two sub-periods, 1990-1995, and 1996-2002. The privatizations of the public-owned banks are concentrated in the second sub-period.

Variable	Average	Standard Deviation
Output*		
1990 to 2002 (4444 obs.)	5,001	18,511
1990 to 1995 (1998 obs.)	3,629	15,135
1996 to 2002 (2446 obs.)	6,116	20,791
Employees		
1990 to 2002	3,261	12,381
1990 to 1995	3,596	13,248
1996 to 2002	2,988	11,625
Capital*		
1990 to 2002	123.9	648.2
1990 to 1995	141.0	789.6
1996 to 2002	110.0	504.7
Other intermediate inputs*		
1990 to 2002	28.0	101.8
1990 to 1995	26.2	92.9
1996 to 2002	28.4	108.4

TABLE	6 -	Sample	summary
-------	-----	--------	---------

* in million of Reais (December 2002 values)

The dispersion of the variables across banks is very large, with the standard errors being nearly four times larger than the mean values. There is no discernible difference between the two sub-periods in this regard. Firm bank heterogeneity is therefore very significant in the sample and for both sub-periods.

Mean production increases between the two sub-periods, alongside concomitant reductions in the use of labor and capital. Intermediate inputs, by contrast, have increased between the two sub-periods.

Figures 1 and 2 show the evolution of the mean values of the variables across time.



Figure 1: Evolution of average output, and number of employees

Figure 2: Evolution of average capital stock, and intermediate inputs



One can then observe that bank output starts to continuously increase since 1993, registering an accumulated growth of 122% between Jun 1993 and December 2002. Employment reduces over the sample period with the largest fall occurring between 1990 and mid-1997. After 2000, employment has recovered a little. Capital stock shows a decreasing trend starting in 1993. Average capital stock halved its value between the peak of December 1992 and December 2002. Intermediate inputs are the only inputs that followed the increasing pattern for the output in the most recent years.

5. Estimation of bank productivity

This section implements Levinsohn and Petrin (2003) methodology to obtain measures of productivity for a sample of Brazilian banks. In the next section, the banklevel productivity measures so obtained are regressed on a number of control variables, including, among them, dummy variables representing privatization of state-owned banks.

Levinsohn and Petrin (2003) modify Olley e Pakes (1996) approach in two ways. First, they replace investment by electricity as a proxy variable for productivity. Second, they do not model the exit decision by the bank. For the sample of Brazilian banks, troubled banks stop reporting employment levels to the Central Bank, which preclude us to implement the second step in Olley and Pakes algorithm. As Levinshohn and Petrin argue, we hope that the use of unbalanced panel data helps to reduce the sample selection bias.

The first step of the algorithm involves the estimation of the following "partially linear" equation:

$$y_{it} = \beta_0 + \beta_l l_{it} + \beta_i i_{it} + \beta_c c_{it} + \beta_k k_{it} + h_t (c_{it}, k_{it}) + time_t + \varepsilon_{it}$$
(15)

where y_{it} is the log of bank output, l_{it} is the log of labor, i_{it} is the log of other intermediate inputs, c_{it} is the log of communications, k_{it} is the log of capital, *time*_t is a trend variable, and ε_{it} is the random error term.

The $h_i(c_{ii}, k_{ii})$ function is estimated by means of a polynomial series expansion where terms of up to the fourth degree of c_{ii} and k_{ii} are used. This series expansion is separately estimated for three different sub-periods: from 1990 to June 1994, from December 1994 to 1997, and from 1998 to 2002. In equation (15) labor is considered as a variable factor. We also estimated models where labor is treated as a fixed factor. When this is the case, labor is also incorporated in the polynomial series expansion.

The first step of the estimation allows one to obtain consistent estimates of the variable factor coefficients, β_i and β_i . Once these coefficients are obtained, we compute the term:

$$y_{it}^{p} = y_{it} - \hat{\beta}_{l} l_{it} - \hat{\beta}_{i} i_{it}$$
(16)

This term is then regressed on a polynomial series in (c_{it}, k_{it}) . The fitted value from this regression is denoted $\hat{\varphi}_t(c_{it}, k_{it})$.

In the second step, consistent estimates for β_c and β_k are obtained through non-linear least squares applied to:

$$y_{it}^{p} = \beta_{0} + \beta_{c}c_{it} + \beta_{k}k_{it} + g[\hat{\varphi}_{t-1}(c_{it-1}, k_{it-1}) - \beta_{0} - \beta_{c}c_{it-1} - \beta_{k}k_{it-1}] + \xi_{it} + \varepsilon_{it}$$
(17)

where ξ_{it} is the innovation term in productivity.

Table 7 presents the production function coefficients estimated through the Levinsohn and Petrin (L-P) algorithm, alongside the coefficients obtained through least squares estimation.¹³

¹³ The standard errors of the coefficients for the fixed inputs and for communications in the L-P models were obtained by bootstrap resampling 100 times. There are fewer observations in the L-P models due to the use of lagged terms in the estimation of (17).

Input	Least Squares	L-P (variable labor)	L-P (fixed labor)
In (labor)	-0.0479*	-0.1485***	0.4409***
	(-1.90)	(-3.30)	(4.17)
In (other intermediate)	0.2599***	0.2137***	0.4391***
	(9.03)	(8.21)	(19.74)
In (communications)	0.6162***	0.6119***	0.2047***
	(22.94)	(8.68)	(4.80)
In (capital)	0.0236	0.0515*	0.064
	(1.50)	(1.89)	(1.07)
time	0.0482***	0.0379***	0.0546***
	(20.96)	(6.06)	(8.24)
Observations	4444	4202	4202

 TABLE 7 - Estimates of production function parameters

t-statistic in parentheses

*,**,*** indicate significant at 10, 5, and 1 percent, respectively

The labor coefficient in the least squares estimation is negative, and marginally significant. When labor is treated as a variable factor in the L-P approach, its coefficient becomes even more negative and highly significant this time. The last column of Table 7 shows the estimates of the L-P approach when labor is treated as a fixed factor. This model shows more reasonable values for the estimated coefficients, although the statistical significance of the capital stock coefficient is still low¹⁴. Thus, for the remaining of the analysis, the coefficients shown in the third column are used as the estimates for the bank production function.

As discussed in section 3, it is important that monotonicity with respect to productivity holds for the communications input. If this assumption is violated, we cannot invert this function to express productivity as a function of observable variables. In order to check the validity of this assumption, communications was regressed on productivity and on the fixed inputs (capital and labor). All the estimated coefficients are positive and significant.

 $^{^{14}}$ The results are robust to different cutoff values for the outliers (0.5%, 1%, 2%), and also to different degrees for the polynomial expansion series (fourth and fifth degrees).

Bank-level (log of) total factor productivity is computed as the difference between actual and fitted output, given by:

$$\omega_{it} = y_{it} - \hat{\beta}_{l} l_{it} - \hat{\beta}_{i} i_{t} - \hat{\beta}_{c} c_{it} - \hat{\beta}_{k} k_{it}$$

Aggregate bank productivity is calculated as the weighted bank-level productivity for each period, where the weight is given by the market share of each bank product in each time period. Figure 3 shows the evolution of the aggregate productivity alongside a more standard measure of productivity, namely labor productivity (calculated as the weighted average of $y_{ii} - l_{ii}$), both normalized to one in June 1990.

Figure 3: Brazilian banking system: total factor, and labor productivity



(June 1990 = 1)

The two aggregate productivity measures display similar temporal patterns although their numerical scales differ. It is interesting to notice that labor productivity underestimates total factor productivity in banking, which is the opposite of what is found for the manufacturing sector in Brazil [Muendler (2002), Schor (2003)]. One possible reason for these differences is the large fall in the capital stock in banking

shown in Figure 3, whereas capital stock has increased in the manufacturing sectors over the 1990's.

Aggregate total factor productivity increases up to June 1997, remaining fairly flat after it. The accumulated productivity growth over the entire period is 13.48%, or an average annual growth rate of 1.02%. There is great heterogeneity across sub-periods though. In the period from June 1990 to June 1997, the average annual growth rate reached 3.70%. From June 1997 to December 2002, there is actually a small fall in the aggregate productivity: there is an accumulated fall of 3.63% or an average annual fall of 0.67%.

Nakane (1999) found similar results through the estimation of cost functions for Brazilian banks accounting for efficiency according to the stochastic frontier methodology¹⁵. According to his estimation, the average cost efficiency index for Brazilian banks reached 0.4151 in the June 1990 to June 1994 period, and increased to 0.5098 in the June 1994 to June 1997 period. Bevilaqua and Loyo (1998) also document cost efficiency gains for a panel of 38 Brazilian banks during the last quarter of 1994 and the second quarter of 1998.

6. Determinants of bank productivity

This section attempts to study the determinants of bank productivity. Special attention is paid to the role of the ownership structure. In face of an environment whereby different corporate changes are affecting the industry, Berger *et al.* (2003) argue that it is important to control for as many of the changes as possible. In other terms, even if the primary interest of the paper relies on the effects of bank privatization, the introduction of controls for corporate changes that do not involve state-owned banks (e.g. domestic M&As or foreign acquisition of domestic banks) are overdue.

¹⁵ See Berger and Humphrey (1997) for a survey of efficiency studies applied to banking.

Moreover, Berger *et al.* (2003) develop a framework where static, selection, and dynamic effects are contemplated. Static effects refer to the differences in performance for groups of banks that have not been involved in any corporate change. Selection effects are those related to the performance differentials for the groups of banks that were involved in some ownership change. Finally, dynamic effects capture the changes in performance for the last group of banks that are *due to* the change in ownership. This framework has been applied to study the Argentinean case by Berger *et al.* (2003), the Brazilian case by Beck, Crivelli, and Summerhill (2003), and the Nigerian case by Beck, Cull, and Jerome (2003).

The Brazilian case seems to be well suited for an application of this methodology. First, many corporate changes affected the banking industry in the 1990's, involving bank privatization, domestic M&As, foreign acquisition of domestic banks, and bank closures. Second, as documented in section 2, the process of bank privatization was very rich in Brazil. Public banks owned by states were offered different solutions to their banks, including, liquidation, outright privatization, federalization followed by privatization, and restructuring.

Static dummy variables were created for those banks that did not face any ownership change over the sample period and were still active by the end of the sample period. Three static dummy variables were created for state-owned banks (*dstatic_state*), for foreign-controlled banks (*dstatic_foreign*), and for branches of foreign banks (*dstatic_branchforeign*). These dummy variables take the value one for the corresponding bank for all the time periods. Domestic private banks are the excluded reference group.

Out of the 242 banks with observations in the sample, 112 (46.28%) of them were active by December 2002 without experiencing any corporate change. Five (2.07% of the total) of them were state-owned banks, 64 (26.45%) of them were domestic private banks, 32 (13.22%) of them were foreign controlled banks, and 11 (4.55%) of them were branches of foreign banks.

The relevance of each group of banks cannot be entirely gauged by their respective numbers due to the presence of many small banks. We therefore computed the market share of each group in December 2002. On this account, the group of

commercial banks that have not experienced any corporate change responded for 79.94% of the market share. The market shares accounted by state-owned banks, domestic private banks, foreign controlled banks, and branches of foreign banks are, respectively, 32.52%, 28.25%, 15.91%, and 3.26%.

Selection dummy variables were created for those banks that have faced some corporate change over the sample period. Eight selection dummy variables were created for state-owned banks that were privatized and acquired by domestic banks (dselection_privatized_domestic), for state-owned banks that were privatized and acquired by foreign banks (dselection_privatized_foreign), for state-owned banks that were first federalized, later privatized and acquired by domestic banks (dselection_federalized_privatized_domestic), for state-owned banks that were first federalized, later privatized acquired by foreign and banks (*dselection_federalized_privatized_foreign*), for state-owned banks that were federalized and not privatized yet (dselection_federalized_notprivatized), for statewere restructured and kept under owned banks that state ownership (dselection_restructured), for domestic banks acquired by other domestic banks (dselection domestic), and for domestic banks acquired by foreign banks (dselection_foreign). The selection dummy variables take the value one for the corresponding banks during all the time periods. Notice that the selection variables control not only for different solutions given to the state-owned banks but also to the ownership of the acquiring bank.¹⁶

Out of the 242 commercial banks in the sample, 55 (22.73%) of them have had some form of control change. State-owned banks account for 19 (7.85% of the total) of the cases with 5 privatized banks bought by domestic banks, 2 privatized banks bought by foreign banks, 2 federalized and privatized banks bought by domestic banks, 1 federalized and privatized bank bought by a foreign bank, 4 federalized and not yet privatized banks, and 5 reestructured state-owned banks. Of the remaining 36 cases (14.88% of the total) involving private banks, half of them are cases of domestic banks merging with other domestic banks whereas the other half are situations where domestic banks are being acquired by foreign ones. In terms of market share, the group of banks that were involved in any form of corporate change accounted for 20.06% of the market in December 2002. The market shares of the state-owned banks, of the domestic banks acquired by other domestic banks, and of the domestic banks acquired by foreign banks are, respectively, of 9.51%, 5.17%, and 5.37%.

Dynamic dummy variables were created for those banks for which the selection dummies were equal to one to date the precise moment when the ownership change occurred. Ten dynamic dummy variables were created for state-owned banks that were privatized and acquired by domestic banks (ddynamic privatized domestic), for stateowned banks that were privatized and acquired by foreign banks (*ddynamic_privatized_foreign*), for state-owned banks that were first federalized, later privatized and acquired by domestic banks dating the time of federalization (*ddynamic_federalized_privatized_domestic_datefederalization*), for state-owned banks that were first federalized, later privatized and acquired by domestic banks dating the time of privatization (*ddynamic_federalized_privatized_domestic_dateprivatization*), for state-owned banks that were first federalized, later privatized and acquired by the time of federalization foreign banks dating (*ddynamic_federalized_privatized_foreign_datefederalization*), for state-owned banks that were first federalized, later privatized and acquired by foreign banks dating the time of privatization (*ddynamic_federalized_privatized_foreign_dateprivatization*), for stateowned banks that were federalized and not privatized yet dating the time of federalization (*ddynamic_federalized_notprivatized*), for state-owned banks that were restructured and kept under state ownership dating the time of restructuring (ddynamic_restructured), for domestic banks acquired by other domestic banks (ddynamic_domestic), and for domestic banks acquired by foreign banks (ddynamic_foreign). The dynamic dummy variables take the value one for the corresponding banks for all the time periods following a certain intervention.

The dynamic dummy variables capture the once-and-for-all changes associated to the interventions. However, in addition to this level effect, the interventions can have differentiated impacts over time. We therefore also created variables measuring the time lapsed since the intervention. Since we use 6-month observations in our sample, such variables are measured in semesters. Ten time variables were created, one for each dynamic dummy variable. The labels for such variables follow the same pattern as the

¹⁶ For some few banks, there was more than one change of control. In such cases, we followed Berger et

ones defined for the dynamic dummy variables with *time* replacing *ddynamic*. For example, *time_federalized_privatized_foreign_datefederalization* measures the time since a state-owned bank that was federalized, privatized and acquired by a foreign bank was federalized. Typically, the *time* variables take the value one in the semester when the intervention occurred, the value two in the following semester, and so on.

We follow Berger *et al.* (2003) and actually exclude from the sample all observations for which the *time* variables equal one. In other terms, the semester during which the intervention occurred is not considered in the sample. The reason for this treatment is to try to control for noise introduced during the event of intervention, which usually produces some discontinuities in previous policies, involves legal costs associated to the intervention, etc.

In addition to the static, selection, and dynamic variables we created another group of variables to deal with the banks that exited the market.¹⁷ Typically, banks have left the market either because they were liquidated or because they changed their activities from commercial banking into something else. Accordingly, we defined five exit dummy variables for liquidated state-owned banks (*dexit_liquidated_state*), for liquidated private banks (*dexit_liquidated_private*), for commercial banking institutions, e.g. investment bank (*dexit_other_bank*), for commercial banks that became other non-banking financial institutions, e.g. broker house (*dexit_other_finance*), and for commercial banks that became other non-financial institutions (*dexit_nonfinance*). The exit dummy variables take the value one for the corresponding banks during all the periods for which they are present in the sample.

Out of the 242 commercial banks present in our unbalanced sample, 75 (30.99%) of them have exited the market. Six (2.48% of the total) state-owned banks were liquidated, 23 (9.50%) private banks were liquidated, 12 (4.96%) commercial banks became investment banks, 8 (3.31%) commercial banks became non-banking financial institutions, and 26 (10.74%) commercial banks became non-financial institutions. Obviously, the market share of the exited banks is zero by December 2002.

al. (2003)'s procedure and only consider the last change.

¹⁷ Beck, Crivelli, and Summerhill (2003) work with a balanced sample, excluding such banks from their analysis. We believe that keeping these banks in the sample is important to better deal with the sample selection problems discussed in section 3.

In addition to the above mentioned variables we also included two additional control variables given by the lagged market share (*market_share*), as measured by the share of each bank output in the sector output in each period, and lagged bank size given by the number of bank branches (*branches*). Time dummies were also included in the estimated regressions, with the aim of capturing macroeconomic effects not accounted for in the estimation.

Table 8 presents the results when the "time" variables capturing the effects of corporate change over time are not included in the regression. Robust standard errors are computed throughout. The coefficients of the time dummies are not reported to spare space.

Variable	Coefficient	T statistics
dstatic_state	-1.0158***	-10.80
dstatic_foreign	0.2594***	4.72
dstatic_branchforeign	0.4413***	5.85
dexit_liquidated_state	-1.9681***	-18.89
dexit_liquidated_private	-0.5585***	-7.04
dexit_other_bank	-0.2243**	-2.08
dexit_other_finance	-0.3516***	-3.42
dexit_nonfinance	0.5255***	6.81
dselection_privatized_domestic	-1.1891***	-14.08
dselection_privatized_foreign	-1.8522***	-27.26
dselection_federalized_privatized_domestic	-1.7767***	-21.94
dselection_federalized_privatized_foreign	-1.1449***	-7.45
dselection_federalized_notprivatized	-1.7532***	-22.83
dselection_restructured	-1.3633***	-16.16
dselection_domestic	-0.1040	-1.22
dselection_foreign	0.0210	0.26
ddynamic_privatized_domestic	0.3158**	2.50
ddynamic_privatized_foreign	0.8214***	3.94
ddynamic_federalized_privatized_domestic_datefederalization	-0.0724	-0.57
ddynamic_federalized_privatized_domestic_dateprivatization	-0.1127	-0.59
ddynamic_federalized_privatized_foreign_datefederalization	-0.1079	-0.71
ddynamic_federalized_privatized_foreign_dateprivatization	0.1777**	2.02
ddynamic_federalized_notprivatized	-0.0940	-1.04
ddynamic_restructured	-0.2575**	-2.35
ddynamic_domestic	0.4764**	2.18
ddynamic_foreign	0.1531	1.14
market_share	18.5562***	6.99
branches	-0.0018***	-11.86
constant	7.1245***	70.19
Observations R-squared	4147 0.3829	

TABLE 8 - Determinants of Log(TFP)

*,**,*** indicate significant at 10, 5, and 1 percent, respectively

The three static dummy variables are highly significant. The results show that state-owned banks are less productive than the private domestic ones (the reference group). On the other hand, both the foreign-controlled banks and the branches of foreign banks are more productive than the private domestic group.

The five exit dummy variables are significant and one of them is positive. Liquidated banks have lower productivity than those commercial banks that left the market due to a transformation in their activities. State-owned liquidated banks have the poorest performance within this group, followed by the private liquidated banks. Commercial banks that left the market and became non-financial institutions were actually more productive than the private domestic banks (reference group). One possible reason for some banks to leave the market is related to the increasing costs of staying in the bank business related to more stringent prudent regulations as well as to the introduction of the new payment system in the country during 2002.

Five of the selection dummy variables are highly significant and negative. The remaining two selection variables are not significant with one of them having a positive sign. Overall, state-owned banks that underwent some corporate change are less productive than private banks that also faced some corporate change. The selection coefficients are non-significant for the latter group, indicating that their productivity is no different than the productivity of the reference group (domestic private banks).

As for the state-owned banks, there is no clear discernible pattern for the selection variables regarding the type of solution given to them. In other terms, one cannot clearly state that state-owned banks that were privatized, or that were first federalized and later privatized, or that were federalized and not privatized, or that were restructured and kept under state control have better or worse performance compared to each other. It is also not possible to state any firm conclusion related to the ownership of the acquiring bank (i.e. private domestic or foreign control) in terms of the selection of state-owned banks.

The static, exit and selection dummies plus the reference excluded group form a sample partition. One can therefore rank the groups according to the estimated coefficient for each dummy. Table 9 reports the ranking in increasing order (i.e. from the less to the more productive groups). The ordering is a weak one because some of the coefficients are not statistically different from each other. Equality coefficient tests are reported for pairs of adjacent variables and we indicate when the null is rejected.

TABLE 9 - Ranking of groups of banks according to static, exit, andselection variables

Group	F Statistics	P-Value
1. Liquidated state-owned bank	1.03	0.3101
2. Privatized and acquired by foreign	0.64	0.4229
3. Federalized, privatized and acquired by domestic	0.06	0.8134
4. Federalized and not privatized	14.41***	0.0001
5. Reestructured and kept under state control	2.68	0.1017
6. Privatized and acquired by domestic	0.06	0.8067
7. Federalized, privatized and acquired by foreign	0.87	0.3510
8. Active state-owned bank	15.70***	0.0001
9. Liquidated private bank	2.91*	0.0883
10. Transformed into financial non-banking institution	0.81	0.3680
11. Transformed into other banking institution	0.88	0.3482
12. Domestic bank acquired by domestic	1.50	0.2210
13. Active domestic bank	0.07	0.7983
14. Domestic bank acquired by foreign	7.27***	0.0071
15. Active foreign controlled bank	4.96**	0.0260
16. Active branch of foreign bank	0.76	0.3833
17. Transformed into non-financial institution		

*,**,*** indicate significant at 10, 5, and 1 percent, respectively

F Statistics report a test of equality for the coefficient of the corresponding group and the

coefficient for the group in the row immediately below. All the tests have a F(1,4094) distribution.

One striking result from Table 9 is that all the groups involving state-owned banks occupy the lowest positions in the rankings. Even the still active state-owned banks have significantly lower productivity than liquidated private banks¹⁸. Returning to the results of Table 8, only five out of the ten dynamic dummy variables are statistically significant. One of them has a negative coefficient, though, which is contrary to expectations. As a matter of fact, only five of the dynamic variables have the expected positive sign. Overall, the dynamic effects are not large in magnitude.

As in Table 9, we can compute a weak ordering of the groups according to the estimated coefficients of the dynamic variables. Table 10 shows the results. We

¹⁸ Possible reasons for these results could be that such banks may be used as government agents, playing a "social" role. For example, Banco do Brasil and CEF are major players in the concession of loans to the rural and to the low-income housing sector, respectively, usually at subsidized rates. Nonetheless, such conjecture needs a careful analysis, examining and comparing lending policies from such banks with their counterparts.

consider the state-owned banks that were first federalized and later privatized during two moments: the period they remained under federal control, and the period following privatization. The variables we name *...datefederalization* and *...dateprivatization* in Table 8 aim at capturing the dynamic impacts of federalization and of privatization, respectively. Thus, Table 10 reports the rankings of ten distinct groups in increasing order.

Group	F Statistics	P-Value
1. Reestructured and kept under state control	0.44	0.5073
2. Federalized, privatized and acquired by domestic, after privatization	0.00	0.9841
3. Federalized, privatized and acquired by foreign, during federalization	0.01	0.9372
4. Federalized and not privatized	0.02	0.8799
5. Federalized, privatized and acquired by domestic, during federalization	1.66	0.1983
6. Domestic bank acquired by foreign	0.02	0.8763
7. Federalized, privatized and acquired by foreign, after privatization	0.87	0.3512
8. Privatized and acquired by domestic	0.42	0.5182
9. Domestic bank acquired by domestic	1.33	0.2491
10. Privatized and acquired by foreign		

*,**,*** indicate significant at 10, 5, and 1 percent, respectively

F Statistics report a test of equality for the coefficient of the corresponding group and the coefficient for the group in the row immediately below. All the tests have a F(1,4094) distribution.

The first five groups in Table 10 present negative impacts for the dynamic dummy variables. All of them involve state-owned banks. Three of such negative impacts are related to state-owned banks *during* periods of restructuring (either under federal or under state control). This result is not surprising because restructuring usually involves the transfer of troubled assets to the federal government. Since total assets are our measure of output, such banks may be facing a reduction in output while keeping unchanged their level of inputs and therefore reducing their productivity.

When the performance of the state-owned banks is measured after the privatization, the dynamic impacts show a more positive outcome. Three of the four possible groups improved their performance after privatization. Moreover, the way a state-owned bank was privatized seems to matter with better performance related to straight privatization (as opposed to first federalize and then privatize).¹⁹

¹⁹ The tests that the dynamic effects for state-owned banks that were privatized and for state-owned banks that were first federalized and then privatized are equal give F statistics of 8.38 (p-value equal to 0.0038)

Moreover, the ownership of the acquiring bank also seems to matter with better dynamic performance related to the privatized banks acquired by foreigners.²⁰ Interestingly, when the corporate change involved only private banks, domestic banks seem to be associated with better dynamic performance than foreign ones, although the statistical significance for testing the difference between them is very low (F statistics equal to 1.64 with a p-value of 0.2010).

The ordering shown in Table 10 can be misleading because any given value for a dynamic variable may have different quantitative implications for different bank groups. For example, a coefficient of, say, 0.2 for a dynamic variable may represent a tremendous improvement for a privatized state-owned bank and not such a great performance for a private bank. We therefore re-computed the rankings of the bank groups displayed in Table 9 by considering the dynamic effects of Table 10. In other terms we add the coefficients for each selection variable and the corresponding dynamic one, comparing the result with the static and exit variables. Table 11 shows the resulting weak ranking in increasing order. For the state-owned banks that were first federalized and later privatized we show their combined performance after the privatization, i.e. we take into account the impacts of the federalization period.

when the acquiring bank is a foreign one and 3.52 (p-value equal to 0.0607) when the acquiring bank is a domestic one.

 $^{^{20}}$ The statistical significance for such tests are not very high though. The tests that the dynamic effects for privatized state-owned banks acquired by foreigners and for privatized state-owned banks acquired by domestic banks are equal give F statistics of 4.52 (p-value equal to 0.0336) when the state-owned bank was privatized by the state, and 2.09 (p-value equal to 0.1486) when the state-owned bank was first federalized and then privatized.

TABLE 11 - Ranking of groups of banks according to static, exit, and selection plus dynamic variables

Group	Previous Ranking	F Statistics
		0.00
1. Liquidated state-owned bank	1	0.00
2. Federalized, privatized and acquired by domestic	3	0.51
3. Federalized and not privatized	4	8.48***
4. Reestructured and kept under state control	5	42.08***
5. Federalized, privatized and acquired by foreign	7	0.05
6. Privatized and acquired by foreign	2	0.00
7. Active state-owned bank	8	1.03
8. Privatized and acquired by domestic	6	6.29**
9. Liquidated private bank	9	2.91*
10. Transformed into financial non-banking institution	10	0.81
11. Transformed into other banking institution	11	4.34**
12. Active domestic bank	13	2.33
13. Domestic bank acquired by foreign	14	0.52
14. Active foreign controlled bank	15	0.29
15. Domestic bank acquired by domestic	12	0.10
16. Active branch of foreign bank	16	0.76
17. Transformed into non-financial institution	17	

*,**,*** indicate significant at 10, 5, and 1 percent, respectively

F Statistics report a test of equality for the coefficient of the corresponding group and the

coefficient for the group in the row immediately below. All the tests have a F(1,4094) distribution.

It remains true that all the lowest positions are still occupied by former or active state-owned banks. Even the best performing in this group, the ones privatized and acquired by domestic banks, are still (significantly) lagging behind the worst performers in the group of private banks, namely the liquidated private banks.

However, some noticeable improvements have been detected for the state-owned banks that were directly privatized. Privatized state-owned banks that were acquired by foreign banks jumped four positions in the ranking while the ones acquired by domestic banks improved two positions. The former group caught up with the active state-owned banks while the latter leapfrogged this group.

The same good performance is not verified for those privatized state-owned banks that were first federalized. The two bank groups belonging to this category lost positions in the ranking. As already mentioned, such banks faced the negative costs of restructuring, which impaired their productivity. In special, those federalized banks that were privatized and bought by domestic banks had such a poor performance that, even after the privatization, they perform no better than the liquidated state-owned banks.

It is also worth mentioning that the alternative of restructuring the state-owned bank and keeping it under state control does not seem to yield good results. After taking into account the dynamic impacts, this group of banks lost one position in the ranking. On the plus side, restructured state-owned banks have significant greater productivity than those federalized state-owned banks that have not been privatized.

Figure 4 gives a graphical representation of the changes affecting the stateowned banks. It shows the eight groups representing these banks during three hypothetical time periods. Time t shows the distribution previous to the corporate change, as represented by Table 9. Time t+1 shows the impact of the first corporate change, e.g. federalization for those banks that were federalized or privatization for those banks that were privatized straight away. Time t+2 shows the final configuration, as represented by Table 10.



Figure 4: State-owned banks: static, exit, selection, and dynamic effects

Domestic banks that were sold to foreign banks leapfrogged the group of active domestic banks while domestic banks that were sold to other domestic banks significantly improved their performance increasing three positions in the ranking.

Overall, the results suggest that dynamic effects associated both to straight privatization and to corporate control change in the private sector worked towards improving productivity of the involved banks.

With regard to the other control variables included in the regression reported in Table 8, the results indicate a positive effect of lagged market share, and a negative effect of the lagged number of branches on bank productivity. Notice that the use of *lagged* market share helps to control for a possible reverse causality channel whereby more productive banks have an edge to increase their market shares. On the other hand, the negative effect for the number of bank branches may be pointing out to scale diseconomies. The operation of extensive branch networks can impart on productivity if the branches are small and geographically dispersed.

The results reported so far assumed that the dynamic impacts of any corporate change have a "once-and-for-all" nature. A more realistic picture may be represented by a setting whereby the dynamic impacts materialize over time. Table 12 reports the results when the "time" variables capturing the effects over time of the corporate changes are also included in the regression.

Variable	Coefficient	T statistics
dstatic_state	-1.0160***	-10.79
dstatic_foreign	0.2595***	4.71
dstatic_branchforeign	0.4413***	5.85
dexit_liquidated_state	-1.9680***	-18.86
dexit_liquidated_private	-0.5570***	-6.98
dexit_other_bank	-0.2252**	-2.09
dexit_other_finance	-0.3539***	-3.48
dexit_nonfinance	0.5276***	6.78
dselection_privatized_domestic	-1.1895***	-14.06
dselection_privatized_foreign	-1.8525***	-27.22
dselection_federalized_privatized_domestic	-1.7770***	-21.91
dselection_federalized_privatized_foreign	-1.1446***	-7.44
dselection_federalized_notprivatized	-1.7537***	-22.79
dselection_restructured	-1.3636***	-16.14
dselection_domestic	-0.1045	-1.23
dselection_foreign	0.0213	0.26
ddynamic_privatized_domestic	0.0221	0.13
ddynamic_privatized_foreign	-0.35545	-1.47
ddynamic_federalized_privatized_domestic_datefederalization	0.2071	0.50
ddynamic_federalized_privatized_domestic_dateprivatization	-1.1072***	-3.30
ddynamic_federalized_privatized_foreign_datefederalization	0.0909	0.44
ddynamic_federalized_privatized_foreign_dateprivatization	0.1112	0.62
ddynamic_federalized_notprivatized	-0.0447	-0.29
ddynamic_restructured	-0.2367	-1.19
ddynamic_domestic	0.3786	0.70
ddynamic_foreign	0.2387	1.22
time_privatized_domestic	0.0557*	1.89
time_privatized_foreign	0.2140***	6.91
time_federalized_privatized_domestic_datefederalization	-0.8510	-0.85
time_federalized_privatized_domestic_dateprivatization	0.5091***	3.11
time_federalized_privatized_foreign_datefederalization	-0.0575	-1.31
time_federalized_privatized_foreign_dateprivatization	0.1021*	1.66
time_federalized_notprivatized	-0.0120	-0.40
time_restructured	-0.0041	-0.12
time_domestic	0.0209	0.20
time_foreign	-0.0162	-0.61
market_share	18.5432***	6.98
branches	-0.0018***	-11.84
constant	7.1243***	70.08
Observations R-squared	4147 0.3833	

TABLE 12 - Determinants of Log(TFP) with effects over time

*,**,*** indicate significant at 10, 5, and 1 percent, respectively

As expected, there is little change in the coefficients for the static, exit, and selection dummy variables. There is also no change in the coefficients for market share

and number of branches. With respect to the dynamic dummy variables, only one of them remains significant but with a negative sign. Five (out of ten) "time" variables are positive with four of them being statistically significant. None of the negative "time" variables are statistically significant.

More interestingly, all the four significant "time" variables are related to privatized banks. This finding gives support to the idea that productivity improvements after privatization may take some time to materialize. It is also worth mentioning that, unlike the previous findings, there is no clear dominance of one type of privatization over the other when the "over time" impacts are taken into account. There is also no clear dominance with regard to the ownership of the acquiring bank on the dynamic impacts of privatization.

Figure 5 shows the results of a counterfactual experiment where the dynamic effects are allowed to vary over time. In time period one, the state-owned banks are ranked according to their static, exit, and selection coefficients displayed in Table 12. The ordering is exactly the same as shown in Table 9 and in Figure 4. We then computed the number of periods during which a typical bank for each group underwent some corporate change. The number of periods is just an average for the banks within each group. It turns out that the longest time span for which an intervention involving state-owned banks was observed in the sample was 10 semesters for those banks that were first federalized and later privatized and bought by foreign banks. Thus, period two in Figure 5 computes the impact effects of federalization for these banks. Such effects are given by the sum of the coefficients for selection, dynamic (date of federalization), and time (date of federalization) variables for this group of banks. In period three, the coefficient for this group of banks is given by the sum of the coefficients for selection, dynamic (date of federalization), and two times the time (date of federalization) variables. And so on. The procedure to compute the coefficients for the other groups of state-owned banks was the same. Hence, Figure 5 preserves the time lengths typically observed in our sample during which the corporate changes were put in effect.



Figure 6: State-owned banks: static, exit, selection, and dynamic effects over time

The results are similar to those depicted in Figure 4 but with richer dynamic responses. The positive "over time" dynamic impacts of privatization are visible for all four involved groups. This effect is particularly impressive for the directly privatized banks bought by foreigners. Those banks that were directly privatized and bought by domestic banks aimed at leapfrogging the active state-owned banks three periods (one and a half years) after being privatized.

Figure 4 suggested a rather disappointing performance for the banks that were first federalized and later privatized to domestic banks. Even after privatization, their productivity was no better than the productivity of the liquidated state-owned banks. Figure 5 helps to explain what happened to such banks. The "over time" effects of federalization were negative to them and the "on impact" effect of privatization was strongly negative. The "over time" effects of privatization, however, are positive. Actually, this group of banks has the greatest coefficient on the "over time" effects, according to the estimates displayed in Table 12. It turns out that the privatization of these banks is of a more recent vintage. So, their apparent dismal performance is explained by the fact that the positive impacts of privatization have not taken their full effect yet. Giving time to time, our estimates suggest that these banks are catching-up with the other groups.

Summing up, our results suggest that state-owned banks are less productive than their private peers. Privatization of state-owned banks improves productivity. In special, the beneficial effects of privatization are spread out over many periods. In addition, restructuring the state-owned banks and keeping it under state control seems to be a choice that is dominated by privatization when the dynamic effects are considered.

7. Conclusions

In the 1990s, the Brazilian banking sector underwent huge transformations. Following the control of the inflationary process, there was an intense wave of mergers and acquisitions, involving not only domestic agents but also foreign banks. Many stateowned banks were privatized; some of them were closed down. Many troubled private banks also went bust. Improved bank regulation and supervision were also put into action. The purpose of this study was to evaluate how bank productivity was affected by these changes. Particular attention was paid to the effects of the privatization of stateowned banks.

The empirical sections of the paper made use of an unbalanced panel data for 242 commercial banks, observed twice a year, from December 1990 to December 2002. Bank-level productivity measures were obtained as the difference between actual and expected output, where the latter is the fitted value from the estimation of a production function. The estimated production function follows the strategy suggested by Levinsohn and Petrin (2003) to account for endogeneity problems.

In the second stage of the investigation, we tried to evaluate the role of some control variables on the level of the bank productivity. Given the varied nature of corporate changes during the sample period, we follow Berger *et al.* (2003) and try to control for static, selection, and dynamic effects. We also include dummy variables controlling for exited banks. The results show a positive association between productivity and bank market share. It also shows negative effects from the number of

bank branches on productivity. Moreover, state-owned banks seem to be less productive than their private competitors. Bank privatization had positive "over time" impact on productivity. The way a state-owned bank is privatized seems to matter when the "over time" impacts of privatization are not computed. In this setting, straight privatization seems to be a superior strategy than federalization followed by privatization. However, when the impacts of privatization over time are taken into consideration, there is no clear dominance of one form of privatization over the other. There is also no clear dominance in the dynamic performance of privatized banks acquired by foreigners over those acquired by domestic banks and vice versa.

References

- ANDIMA-IBGE (1997). Sistema Financeiro: Uma Análise a Partir das Contas Nacionais 1990-1995. Rio de Janeiro, Andima & IBGE.
- Baer, W., and Nazmi, N. (2000). "Privatization and restructuring of banks in Brazil", *Quarterly Review of Economics and Finance*, **40**, 3-24.
- Beck, T., Crivelli, J. M., and Summerhill, W. (2003). "State bank transformation in Brazil – choices and consequences". Conference on Bank Privatization, World Bank.
- Beck, T., Cull, R., and Jerome, A. (2003). "Bank privatization and performance: empirical evidence from Nigeria". Conference on Bank Privatization, World Bank.
- Berger, A., and Humphrey, D. (1992). "Measurement and efficiency issues in commercial banking", in Griliches, Z. (editor) *Output Measurement in the Service Sectors*. University of Chicago Press, NBER.
- ----- (1997). "Efficiency of financial institutions: international survey and directions for future research", *European Journal of Operational Research*, **98**, 175-212.
- Berger, A., Clarke, G. R. G., Cull, R., Klapper, L., and Udell, G.F. (2003). "Corporate governance and bank performance: a joint analysis of the static, selection, and dynamic effects of domestic, foreign, and state ownership, including domestic M&As, foreign acquisitions, and privatization". Conference on Bank Privatization, World Bank.
- Bevilaqua, A. (2000). "State-government bailouts in Brazil". Department of Economics, PUC, Rio de Janeiro, Discussion Paper no. 421.
- Bevilaqua, A., and Loyo, E. (1998). "Openess and efficiency in Brazilian banking". Department of Economics, PUC, Rio de Janeiro, Discussion Paper no. 390.
- Boehmer, E., Nash, R. C., and Netter, J. M. (2003). "Bank privatization in developing and developed countries: cross-sectional evidence on the impact of economic and political factors". Conference on Bank Privatization, World Bank.
- Bonaccorsi di Patti, E., and Hardy, D. C. (2003). "The effects of banking system reform in Pakistan". Conference on Bank Privatization, World Bank.
- Bonin, J. P., Hasan, I., and Wachtel, P. (2003). "Privatization matters: bank performance in transition countries". Conference on Bank Privatization, World Bank.
- Clarke, G. R. G., Cull, R., and Shirley, M. (2003). "Empirical studies on bank privatization: an overview". Conference on Bank Privatization, World Bank.

- Fixler, D. J., and Zieschang, K. D. (1992). "User costs, shadow prices, and the real output of banks", in Griliches, Z. (editor) *Output Measurement in the Service Sectors*. University of Chicago Press, NBER.
- Haber, S., and Kantor, S. (2003). "Getting privatization wrong: the Mexican banking system, 1991-2003". Conference on Bank Privatization, World Bank.
- Haynes, M., and Thompson, S. (1999). "The productivity effects of bank mergers: evidence from the UK building societies", *Journal of Banking and Finance*, **23**, 825-846.
- Levine, R. (1997). "Financial development and economic growth: views and agenda", *Journal of Economic Literature*, **35**, 688-726.
- Levinsohn, J., and Petrin, A. (1999). "When industries become more productive, do firms? Investigating productivity dynamics", NBER Working Paper No. 6893.
- ----- (2003). "Estimating production functions using inputs to control for unobservables", *Review of Economic Studies*, **70**, 317-342.
- Megginson, W. (2003). "The economics of bank privatization". Conference on Bank Privatization, World Bank.
- Muendler, M.-A. (2002). "Trade, technology, and productivity: a study of Brazilian manufacturers, 1986-1998". Mimeo, University of California, San Diego.
- Nakane, M. I. (1999). "Productive efficiency in the Brazilian banking sector". Mimeo, IPE-USP, Seminar 20/99.
- Ness, Jr., W. L. (2003). "Reducing government bank presence in the Brazilian financial system. Why and how", *Quarterly Review of Economics and Finance*, **40**, 71-84.
- Nguyen, N., and Williams, J. (2003). "Liberalisation, ownership and efficiency issues: a comparative study of South East Asian banking". Conference on Bank Privatization, World Bank.
- Olley, G., and Pakes, A. (1996). "The dynamics of productivity in the telecommunications equipment industry", *Econometrica*, **64**, 1263-1297.
- Omran, M. (2003). "Privatization, state ownership, and the performance of Egyptian banks". Conference on Bank Privatization, World Bank.
- Otchere, I. (2003). "Do privatized banks in middle- and low-income countries perform better than rival banks? An intra-industry analysis of bank privatization". Conference on Bank Privatization, World Bank.
- Schor, A. (2003). "Productivity: competition, embodied technology and heterogeneous response to tariff reduction – evidence from Brazilian manufacturing firms". Mimeo, Fundação Getúlio Vargas, São Paulo.
- Werlang, S. R. da C., and Fraga Neto, A. (1995). "Os bancos estaduais e o descontrole fiscal: alguns aspectos", *Revista Brasileira de Economia*, **49**, 265-275.

Appendix 1: State-owned banks in Brazil

State-owned banks in Brazil operate under two government level structures: at the federal, and at the state (province) level. There were 32 state-owned commercial banks operating in Brazil in December 1994, which were reduced to 16 by December 2002. Eleven banks were privatized, four banks are currently under Central Bank intervention for future privatization, and ten commercial banks were/are being closed down.

Table 13 shows the list of the privatized banks, which includes banks for the following Brazilian states: Amazonas (AM), Bahia (BA), Goiás (GO), Minas Gerais (MG), Paraíba (PB), Paraná (PR), Pernambuco (PE), Rio de Janeiro (RJ), and São Paulo (SP). In addition, a bank previously owned by the federal government (Banco Meridional) was also privatized.²¹

Banespa, BEA, and BEG are cases of state-owned banks that were federalized previous to privatization. The other banks displayed in Table 13 were directly privatized.

The four commercial banks that are being prepared for the Central Bank for future privatization include the banks that were previously owned by the following states: Ceará (CE, BEC), Maranhão (MA, BEM), Piauí (PI, BEP), and Santa Catarina (SC, BESC).²²

²¹ Meridional was later sold to Santander in 2000. For reasons explained in section 6, this bank is therefore not included as a privatized state-owned bank.

²² BEM was privatized and bought by Bradesco on February 10, 2004.

Bank	Privatization Year	Buyer
Banerj (RJ)	1997	Itaú
Meridional (federal)	1997	Bozano Simonsen
Bandepe (PE)	1998	ABN Amro
Bemge (MG)	1998	Itaú
Credireal (MG)	1998	Bradesco
Baneb (BA)	1999	Bradesco
Banespa (SP)	2000	Santander
Banestado (PR)	2000	Itaú
Paraiban (PB)	2001	ABN Amro
BEA (AM)	2002	Bradesco
BEG (GO)	2002	Itaú

TABLE 13 - Privatized Banks

Banks that were previously owned by the following states were closed down: Acre (AC, Banacre), Alagoas (AL, Produban), Amapá (AP, Banap), Goiás (GO, Caixego), Mato Grosso (MT, Bemat), Minas Gerais (MG, Minascaixa), Rio de Janeiro (RJ, Bancoerj), Rio Grande do Norte (RN, Bandern), Roraima (RR, Baner), and Rondônia (RO, Beron).

Appendix 2: Data Related Issues

A. Variables in the production function

The dependent variable in the production function is the value of the bank working assets. Working assets is total assets less fixed assets.

Capital stock includes premises, equipment, other fixed assets, and rented/leased premises and equipment. Capital stock is net of depreciation. Since capital stock is taken as a fixed factor in the production function, we lagged it one period in the estimation.

Intermediate inputs are the sum of communications costs, water, electricity, and gas bill costs, maintenance costs, non-durable goods acquisition costs, data processing costs, and transport costs. The proxy variable for productivity is communications costs.

The source for all the previous variables is COSIF, a monthly report on balance sheet and income statements accounts that all commercial banks in Brazil have to send to the Brazilian Central Bank.

The number of employees is the only variable that does not come from COSIF. The source of this variable is still the Brazilian Central Bank (DEFIN/DINFO unit).

Nominal values from December 1990 to June 1994 were first transformed into *Reais*, the new currency in effect from July 1994 onwards (1 *Real* = 2750 *Cruzeiros*). Constant values were obtained deflating the nominal values through the evolution of the IGP-DI, the general price index calculated by Fundação Getúlio Vargas (FGV). All constant values were converted to December 2002 values.

B. Explanatory variables in the productivity equation

Market share is the ratio of a bank output to the total output in each time period.

The Brazilian Central Bank (DEFIN/DINFO unit) provided the information on the number of bank branches.

The dummy variables for ownership control and transfer were formed from tables elaborated by the Brazilian Central Bank (DEORF unit) and available online. The only information not readily available was the restructuring date for the banks that kept their state ownership. Thorsten Beck kindly sent us this information, which was collected from different sources, including newspapers.

Banco Central do Brasil

Trabalhos para Discussão

Os Trabalhos para Discussão podem ser acessados na internet, no formato PDF, no endereço: http://www.bc.gov.br

Working Paper Series

Working Papers in PDF format can be downloaded from: http://www.bc.gov.br

1	Implementing Inflation Targeting in Brazil Joel Bogdanski, Alexandre Antonio Tombini and Sérgio Ribeiro da Costa Werlang	Jul/2000
2	Política Monetária e Supervisão do Sistema Financeiro Nacional no Banco Central do Brasil Eduardo Lundberg	Jul/2000
	Monetary Policy and Banking Supervision Functions on the Central Bank <i>Eduardo Lundberg</i>	Jul/2000
3	Private Sector Participation: a Theoretical Justification of the Brazilian Position <i>Sérgio Ribeiro da Costa Werlang</i>	Jul/2000
4	An Information Theory Approach to the Aggregation of Log-Linear Models <i>Pedro H. Albuquerque</i>	Jul/2000
5	The Pass-Through from Depreciation to Inflation: a Panel Study Ilan Goldfajn and Sérgio Ribeiro da Costa Werlang	Jul/2000
6	Optimal Interest Rate Rules in Inflation Targeting Frameworks José Alvaro Rodrigues Neto, Fabio Araújo and Marta Baltar J. Moreira	Jul/2000
7	Leading Indicators of Inflation for Brazil Marcelle Chauvet	Sep/2000
8	The Correlation Matrix of the Brazilian Central Bank's Standard Model for Interest Rate Market Risk José Alvaro Rodrigues Neto	Sep/2000
9	Estimating Exchange Market Pressure and Intervention Activity Emanuel-Werner Kohlscheen	Nov/2000
10	Análise do Financiamento Externo a uma Pequena Economia Aplicação da Teoria do Prêmio Monetário ao Caso Brasileiro: 1991–1998 Carlos Hamilton Vasconcelos Araújo e Renato Galvão Flôres Júnior	Mar/2001
11	A Note on the Efficient Estimation of Inflation in Brazil <i>Michael F. Bryan and Stephen G. Cecchetti</i>	Mar/2001
12	A Test of Competition in Brazilian Banking Márcio I. Nakane	Mar/2001

13	Modelos de Previsão de Insolvência Bancária no Brasil Marcio Magalhães Janot	Mar/2001
14	Evaluating Core Inflation Measures for Brazil <i>Francisco Marcos Rodrigues Figueiredo</i>	Mar/2001
15	Is It Worth Tracking Dollar/Real Implied Volatility? Sandro Canesso de Andrade and Benjamin Miranda Tabak	Mar/2001
16	Avaliação das Projeções do Modelo Estrutural do Banco Central do Brasil para a Taxa de Variação do IPCA Sergio Afonso Lago Alves	Mar/2001
	Evaluation of the Central Bank of Brazil Structural Model's Inflation Forecasts in an Inflation Targeting Framework <i>Sergio Afonso Lago Alves</i>	Jul/2001
17	Estimando o Produto Potencial Brasileiro: uma Abordagem de Função de Produção <i>Tito Nícias Teixeira da Silva Filho</i>	Abr/2001
	Estimating Brazilian Potential Output: a Production Function Approach <i>Tito Nícias Teixeira da Silva Filho</i>	Aug/2002
18	A Simple Model for Inflation Targeting in Brazil Paulo Springer de Freitas and Marcelo Kfoury Muinhos	Apr/2001
19	Uncovered Interest Parity with Fundamentals: a Brazilian Exchange Rate Forecast Model <i>Marcelo Kfoury Muinhos, Paulo Springer de Freitas and Fabio Araújo</i>	May/2001
20	Credit Channel without the LM Curve Victorio Y. T. Chu and Márcio I. Nakane	May/2001
21	Os Impactos Econômicos da CPMF: Teoria e Evidência <i>Pedro H. Albuquerque</i>	Jun/2001
22	Decentralized Portfolio Management Paulo Coutinho and Benjamin Miranda Tabak	Jun/2001
23	Os Efeitos da CPMF sobre a Intermediação Financeira Sérgio Mikio Koyama e Márcio I. Nakane	Jul/2001
24	Inflation Targeting in Brazil: Shocks, Backward-Looking Prices, and IMF Conditionality Joel Bogdanski, Paulo Springer de Freitas, Ilan Goldfajn and Alexandre Antonio Tombini	Aug/2001
25	Inflation Targeting in Brazil: Reviewing Two Years of Monetary Policy 1999/00 Pedro Fachada	Aug/2001
26	Inflation Targeting in an Open Financially Integrated Emerging Economy: the Case of Brazil Marcelo Kfoury Muinhos	Aug/2001

27	Complementaridade e Fungibilidade dos Fluxos de Capitais Internacionais Carlos Hamilton Vasconcelos Araújo e Renato Galvão Flôres Júnior	Set/2001
28	Regras Monetárias e Dinâmica Macroeconômica no Brasil: uma Abordagem de Expectativas Racionais <i>Marco Antonio Bonomo e Ricardo D. Brito</i>	Nov/2001
29	Using a Money Demand Model to Evaluate Monetary Policies in Brazil Pedro H. Albuquerque and Solange Gouvêa	Nov/2001
30	Testing the Expectations Hypothesis in the Brazilian Term Structure of Interest Rates Benjamin Miranda Tabak and Sandro Canesso de Andrade	Nov/2001
31	Algumas Considerações sobre a Sazonalidade no IPCA Francisco Marcos R. Figueiredo e Roberta Blass Staub	Nov/2001
32	Crises Cambiais e Ataques Especulativos no Brasil <i>Mauro Costa Miranda</i>	Nov/2001
33	Monetary Policy and Inflation in Brazil (1975-2000): a VAR Estimation André Minella	Nov/2001
34	Constrained Discretion and Collective Action Problems: Reflections on the Resolution of International Financial Crises <i>Arminio Fraga and Daniel Luiz Gleizer</i>	Nov/2001
35	Uma Definição Operacional de Estabilidade de Preços <i>Tito Nícias Teixeira da Silva Filho</i>	Dez/2001
36	Can Emerging Markets Float? Should They Inflation Target? <i>Barry Eichengreen</i>	Feb/2002
37	Monetary Policy in Brazil: Remarks on the Inflation Targeting Regime, Public Debt Management and Open Market Operations Luiz Fernando Figueiredo, Pedro Fachada and Sérgio Goldenstein	Mar/2002
38	Volatilidade Implícita e Antecipação de Eventos de <i>Stress</i> : um Teste para o Mercado Brasileiro <i>Frederico Pechir Gomes</i>	Mar/2002
39	Opções sobre Dólar Comercial e Expectativas a Respeito do Comportamento da Taxa de Câmbio <i>Paulo Castor de Castro</i>	Mar/2002
40	Speculative Attacks on Debts, Dollarization and Optimum Currency Areas Aloisio Araujo and Márcia Leon	Apr/2002
41	Mudanças de Regime no Câmbio Brasileiro Carlos Hamilton V. Araújo e Getúlio B. da Silveira Filho	Jun/2002
42	Modelo Estrutural com Setor Externo: Endogenização do Prêmio de Risco e do Câmbio Marcelo Kfoury Muinhos, Sérgio Afonso Lago Alves e Gil Riella	Jun/2002

43	The Effects of the Brazilian ADRs Program on Domestic Market Efficiency Benjamin Miranda Tabak and Eduardo José Araújo Lima	Jun/2002
44	Estrutura Competitiva, Produtividade Industrial e Liberação Comercial no Brasil <i>Pedro Cavalcanti Ferreira e Osmani Teixeira de Carvalho Guillén</i>	Jun/2002
45	Optimal Monetary Policy, Gains from Commitment, and Inflation Persistence <i>André Minella</i>	Aug/2002
46	The Determinants of Bank Interest Spread in Brazil Tarsila Segalla Afanasieff, Priscilla Maria Villa Lhacer and Márcio I. Nakane	Aug/2002
47	Indicadores Derivados de Agregados Monetários Fernando de Aquino Fonseca Neto e José Albuquerque Júnior	Set/2002
48	Should Government Smooth Exchange Rate Risk? Ilan Goldfajn and Marcos Antonio Silveira	Sep/2002
49	Desenvolvimento do Sistema Financeiro e Crescimento Econômico no Brasil: Evidências de Causalidade <i>Orlando Carneiro de Matos</i>	Set/2002
50	Macroeconomic Coordination and Inflation Targeting in a Two- Country Model Eui Jung Chang, Marcelo Kfoury Muinhos and Joanílio Rodolpho Teixeira	Sep/2002
51	Credit Channel with Sovereign Credit Risk: an Empirical Test <i>Victorio Yi Tson Chu</i>	Sep/2002
52	Generalized Hyperbolic Distributions and Brazilian Data José Fajardo and Aquiles Farias	Sep/2002
53	Inflation Targeting in Brazil: Lessons and Challenges André Minella, Paulo Springer de Freitas, Ilan Goldfajn and Marcelo Kfoury Muinhos	Nov/2002
54	Stock Returns and Volatility Benjamin Miranda Tabak and Solange Maria Guerra	Nov/2002
55	Componentes de Curto e Longo Prazo das Taxas de Juros no Brasil Carlos Hamilton Vasconcelos Araújo e Osmani Teixeira de Carvalho de Guillén	Nov/2002
56	Causality and Cointegration in Stock Markets: the Case of Latin America Benjamin Miranda Tabak and Eduardo José Araújo Lima	Dec/2002
57	As Leis de Falência: uma Abordagem Econômica Aloisio Araujo	Dez/2002
58	The Random Walk Hypothesis and the Behavior of Foreign Capital Portfolio Flows: the Brazilian Stock Market Case <i>Benjamin Miranda Tabak</i>	Dec/2002
59	Os Preços Administrados e a Inflação no Brasil Francisco Marcos R. Figueiredo e Thaís Porto Ferreira	Dez/2002

60	Delegated Portfolio Management Paulo Coutinho and Benjamin Miranda Tabak	Dec/2002
61	O Uso de Dados de Alta Freqüência na Estimação da Volatilidade e do Valor em Risco para o Ibovespa João Maurício de Souza Moreira e Eduardo Facó Lemgruber	Dez/2002
62	Taxa de Juros e Concentração Bancária no Brasil Eduardo Kiyoshi Tonooka e Sérgio Mikio Koyama	Fev/2003
63	Optimal Monetary Rules: the Case of Brazil Charles Lima de Almeida, Marco Aurélio Peres, Geraldo da Silva e Souza and Benjamin Miranda Tabak	Feb/2003
64	Medium-Size Macroeconomic Model for the Brazilian Economy Marcelo Kfoury Muinhos and Sergio Afonso Lago Alves	Feb/2003
65	On the Information Content of Oil Future Prices <i>Benjamin Miranda Tabak</i>	Feb/2003
66	A Taxa de Juros de Equilíbrio: uma Abordagem Múltipla Pedro Calhman de Miranda e Marcelo Kfoury Muinhos	Fev/2003
67	Avaliação de Métodos de Cálculo de Exigência de Capital para Risco de Mercado de Carteiras de Ações no Brasil Gustavo S. Araújo, João Maurício S. Moreira e Ricardo S. Maia Clemente	Fev/2003
68	Real Balances in the Utility Function: Evidence for Brazil Leonardo Soriano de Alencar and Márcio I. Nakane	Feb/2003
69	r-filters: a Hodrick-Prescott Filter Generalization Fabio Araújo, Marta Baltar Moreira Areosa and José Alvaro Rodrigues Neto	Feb/2003
70	Monetary Policy Surprises and the Brazilian Term Structure of Interest Rates <i>Benjamin Miranda Tabak</i>	Feb/2003
71	On Shadow-Prices of Banks in Real-Time Gross Settlement Systems <i>Rodrigo Penaloza</i>	Apr/2003
72	O Prêmio pela Maturidade na Estrutura a Termo das Taxas de Juros Brasileiras <i>Ricardo Dias de Oliveira Brito, Angelo J. Mont'Alverne Duarte e Osmani</i> <i>Teixeira de C. Guillen</i>	Maio/2003
73	Análise de Componentes Principais de Dados Funcionais – Uma Aplicação às Estruturas a Termo de Taxas de Juros Getúlio Borges da Silveira e Octavio Bessada	Maio/2003
74	Aplicação do Modelo de Black, Derman & Toy à Precificação de Opções Sobre Títulos de Renda Fixa Octavio Manuel Bessada Lion, Carlos Alberto Nunes Cosenza e César das Neves	Maio/2003
75	Brazil's Financial System: Resilience to Shocks, no Currency Substitution, but Struggling to Promote Growth <i>Ilan Goldfajn, Katherine Hennings and Helio Mori</i>	Jun/2003

76	Inflation Targeting in Emerging Market Economies Arminio Fraga, Ilan Goldfajn and André Minella	Jun/2003
77	Inflation Targeting in Brazil: Constructing Credibility under Exchange Rate Volatility André Minella, Paulo Springer de Freitas, Ilan Goldfajn and Marcelo Kfoury Muinhos	Jul/2003
78	Contornando os Pressupostos de Black & Scholes: Aplicação do Modelo de Precificação de Opções de Duan no Mercado Brasileiro <i>Gustavo Silva Araújo, Claudio Henrique da Silveira Barbedo, Antonio</i> <i>Carlos Figueiredo, Eduardo Facó Lemgruber</i>	Out/2003
79	Inclusão do Decaimento Temporal na Metodologia Delta-Gama para o Cálculo do VaR de Carteiras Compradas em Opções no Brasil Claudio Henrique da Silveira Barbedo, Gustavo Silva Araújo, Eduardo Facó Lemgruber	Out/2003
80	Diferenças e Semelhanças entre Países da América Latina: uma Análise de <i>Markov Switching</i> para os Ciclos Econômicos de Brasil e Argentina Arnildo da Silva Correa	Out/2003
81	Bank Competition, Agency Costs and the Performance of the Monetary Policy <i>Leonardo Soriano de Alencar and Márcio I. Nakane</i>	Jan/2004
82	Carteiras de Opções: Avaliação de Metodologias de Exigência de Capital no Mercado Brasileiro <i>Cláudio Henrique da Silveira Barbedo e Gustavo Silva Araújo</i>	Mar/2004
83	Does Inflation Targeting Reduce Inflation? An Analysis for the OECD Industrial Countries <i>Thomas Y. Wu</i>	May/2004
84	Speculative Attacks on Debts and Optimum Currency Area: A Welfare Analysis Aloisio Araujo and Marcia Leon	May/2004
85	Risk Premia for Emerging Markets Bonds: Evidence from Brazilian Government Debt, 1996-2002 <i>André Soares Loureiro and Fernando de Holanda Barbosa</i>	May/2004
86	Identificação do Fator Estocástico de Descontos e Algumas Implicações sobre Testes de Modelos de Consumo Fabio Araujo e João Victor Issler	Maio/2004
87	Mercado de Crédito: uma Análise Econométrica dos Volumes de Crédito Total e Habitacional no Brasil Ana Carla Abrão Costa	Dez/2004
88	Ciclos Internacionais de Negócios: uma Análise de Mudança de Regime Markoviano para Brasil, Argentina e Estados Unidos <i>Arnildo da Silva Correa e Ronald Otto Hillbrecht</i>	Dez/2004
89	O Mercado de <i>Hedge</i> Cambial no Brasil: Reação das Instituições Financeiras a Intervenções do Banco Central Fernando N. de Oliveira	Dez/2004